

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL089056

Key Points:

- Anthropogenic black carbon (BC) thins cirrus clouds by inhibiting homogeneous nucleation, inducing moderate global cooling
- The cooling impact saturates at high BC number and nucleating efficiency, due to cirrus thickening via enhanced heterogeneous ice formation
- BC cooling by cirrus thinning requires colocation of the aerosol with abundant, medium-thickness homogeneously formed cirrus

Supporting Information:

Supporting Information S1

Correspondence to:

Z. McGraw, zachary.mcgraw@geo.uio.no

Citation:

McGraw, Z., Storelvmo, T., Samset, B. H., & Stjern, C. W. (2020). Global radiative impacts of black carbon acting as ice nucleating particles. *Geophysical Research Letters*, 47, e2020GL089056. https://doi.org/10.1029/2020GL089056

Received 28 MAY 2020 Accepted 29 SEP 2020 Accepted article online 7 OCT 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Global Radiative Impacts of Black Carbon Acting as Ice Nucleating Particles

Zachary McGraw¹, Trude Storelvmo¹, Bjørn Hallvard Samset², and Camilla Weum Stjern²

¹Department of Geosciences, University of Oslo, Oslo, Norway, ²Center for International Climate and Environmental Research-Oslo (CICERO), Oslo, Norway

Abstract Black carbon (BC) aerosols from incomplete combustion generally warm the climate, but the magnitudes of their various interactions with climate are still uncertain. A key knowledge gap is their role as ice nucleating particles (INPs), enabling ice formation in clouds. Here we assess the global radiative impacts of BC acting as INPs, using simulations with the Community Earth System Model 2 climate model updated to include new laboratory-based ice nucleation parameterizations. Overall, we find a moderate cooling through changes to stratiform cirrus clouds, counteracting the well-known net warming from BC's direct scattering and absorption of radiation. Our best estimates indicate that BC INPs generally thin cirrus by indirectly inhibiting the freezing of solution aerosol, with a global net radiative impact of $-0.13 \pm 0.07 \text{ W/m}^2$. Sensitivity tests of BC amounts and ice nucleating efficiencies, and uncertainties in the environment where ice crystals form, show a potential range of impacts from -0.30 to +0.02 W/m².

Plain Language Summary Airborne black carbon particles generally warm Earth's surface and lower atmosphere, but the individual mechanisms that contribute to and offset this effect are difficult to measure and poorly understand. Highly uncertain is the impact of black carbon's ability to enable ice to form in clouds. Here we use simulations of Earth's climate to estimate the global impact of black carbon's role in ice formation. We find that, in this role, black carbon causes a moderate cooling. This is mostly due to black carbon forming ice in high, thin clouds. Using a mix of simulations to represent the range of plausible circumstances, we calculate bounds on this impact.

1. Introduction

Black carbon (BC) influences Earth's climate through a variety of mechanisms (Bond et al., 2013; Peng et al., 2016), many of which are still poorly constrained. BC has a "very high probability" of warming Earth's climate overall, primarily through its ability to absorb solar radiation (Bond et al., 2013). Recent multimodel studies have consistently quantified the surface warming from early 21st century anthropogenic BC emissions to be around +0.1°C (Baker et al., 2015; Samset et al., 2016; Stjern et al., 2017). However, there are still considerable model differences, linked partly to the low confidence in assessments of BC's diverse roles in clouds (Bond et al., 2013; Stjern et al., 2017). One such mechanism is BC's ability to act as surfaces where ice crystals form at relatively low saturations (ice nucleating particles [INPs]). Ice clouds have an important role in Earth's radiative balance by both reflecting incoming shortwave (SW) radiation back to space and absorbing outgoing longwave (LW) radiation and have overall been found to warm the atmosphere by as much as $5.1 \pm 3.8 \text{ W/m}^2$ globally (Hong et al., 2016). Depending on circumstances, the presence of INPs can either increase or decrease the number density of ice crystals in clouds. Here the impact of BC INPs depends on the number of BC particles at ice cloud altitudes, their ice nucleating efficiency, and their influence on processes that compete for water vapor. Larger (lower) densities of ice crystals typically cause optically thicker (thinner) clouds. INP changes may either strengthen or weaken ice cloud warming effects depending on the balance between LW and SW impacts.

While, in some cases, the presence of INPs will optically thicken ice clouds by enabling crystals to form on their surfaces (*heterogeneous* nucleation), among cirrus that initially form through the freezing of liquid aerosol at high saturations (*homogeneous* freezing/nucleation), the impact of INPs tends to be the opposite (Sullivan et al., 2016; Zhao et al., 2018). Figure 1a shows a schematic of the two mechanisms. INP presence enables water vapor to be consumed as new ice crystals form heterogeneously and grow. By acting as a sink



Geophysical Research Letters



Figure 1. Impacts of anthropogenic black carbon INPs on cirrus clouds and radiation. Shown in counterclockwise order are a schematic of the two competing ice nucleation mechanisms in cirrus (a), simulated anthropogenic BC impacts on heterogeneous and homogeneous cirrus nucleation rates (vertically integrated, in number of events per m^2 per second) (b), impacts on visible shortwave optical depth integrated over high clouds (<400 mb) (c), impacts on top-of-atmosphere cirrus radiative effects (in W/m²) (d), and a comparison between our finding of the total net BC INP effect (dominated by cirrus effects) compared with values from Bond et al. (2013) (e). Shading in (b)–(d) shows the 25th and 75th percentiles among columns within each zonal band. Model averages of these variables are shown in Figure S1. Bounds on our result in (e) are the range of results among sensitivity simulations, while all effects shown are for anthropogenic BC.

of water vapor, these INPs can prevent conditions from reaching the saturations necessary for homogeneous freezing. Due to the greater atmospheric abundance of solution droplets than INPs, INP increases often result in cirrus optical thinning by decreasing the number and increasing the size of ice crystals. Whether INPs cause thinning or thickening of cirrus depends on which of the nucleation mechanisms dominates, which itself depends on cirrus updraft speeds, the presence of INPs both new and prior (Penner et al., 2018), availability of solution droplets for homogeneous freezing, and the type of cirrus affected (since radiative effects vary for instance by cloud top temperature and underlying albedo; Corti & Peter, 2009). Global assessments of INP impacts are limited by inadequacies in global climate models, which typically have insufficient resolutions to resolve the vertical motions that both induce high saturations and bring aerosol to ice cloud altitudes, relying instead on parameterizations. These issues are compounded with difficulties in simulating aerosol sinks, disagreement between laboratory findings of ice nucleating efficiency (Ullrich et al., 2017), and limited aerosol complexity in models (Liu et al., 2016).

BC number concentrations at high altitudes are difficult to assess due to large uncertainties in emissions, transport, and lifetime (Bond et al., 2013; Boucher et al., 2013, 2016; Lund et al., 2018). Laboratory experiments have shown large variation in BC ice nucleating efficiency by particle morphology, size, and aging but have evidenced their potential especially in cirrus clouds (Kanji et al., 2017; Mahrt et al., 2018, 2020; Ullrich et al., 2017). Past model-based estimates of anthropogenic BC INP radiative impacts have ranged from as much as -0.4 to +0.4 W/m² in cirrus to $+0.18 \pm 0.18$ W/m² in mixed-phase clouds (Bond et al., 2013), with a more recent assessments indicating near -0.3 W/m² in cirrus (Penner et al., 2018).

This study assesses the radiative impacts of anthropogenic BC INPs using the atmosphere component of the Community Earth System Model (CESM) v2.0.1 global climate model modified to include laboratory-based parameterizations of BC ice nucleating potential (Ullrich et al., 2017). Simulations were designed to evaluate effective radiative forcings (ERFs), a standard forcing measure that is a useful predictor of global mean near-surface air temperature change (Myhre et al., 2013). We also conducted sensitivity tests to account



for how many of the abovementioned uncertainties may influence BC INP impacts. The result is a comprehensive bounding of the potential climate implications of anthropogenic BC emissions through INP interactions, taking into account the best available knowledge on aerosol-cloud microphysics and Earth System modeling.

2. Methods

Results are based on output from simulations running the CESM2.0.1 (Danabasoglu et al., 2020) model at $1.9^{\circ} \times 2.5^{\circ}$ resolution with 32 vertical levels, using the atmosphere component CAM6 and fixed sea surface temperatures. Simulations used the MAM4 modal aerosol module (Liu et al., 2016), which includes two BC aerosol species, with BC emitted in primary mode and transferred to the accumulation mode as it is aged with coatings of sulfate aerosol. BC emissions were fed in from files containing preindustrial (PI; 1750) and present-day (PD; 2000) mass and number surface fluxes (Hoesly et al., 2018). All simulations included BC emissions from fossil fuel and biofuel sources, as well as changes in biomass burning BC emissions since PI times. In a sensitivity test, we also added aircraft soot, which was not included in either the core simulations or other sensitivity tests.

Ice nucleation processes on BC and mineral dust INPs were replaced with the parameterizations of Ullrich et al. (2017). These parameterizations were derived from laboratory experiments over 11 years that used the Aerosol Interaction and Dynamics chamber at cirrus and mixed-phase temperatures. In Ullrich et al. (2017), heterogeneously formed ice crystal numbers are calculated as functions of temperature, ice saturation ratio, and aerosol surface area density, using the concept of ice nucleation active surface site density (Hoose & Möhler, 2012). Since BC INPs are known to become less active as the aerosol become coated with solution droplets, we scaled down INP efficiency of accumulation mode BC using a 1% ice nucleation active surface site scaling as in Ullrich et al. (2019). To enable heterogeneous nucleation from Ullrich et al. (2017) to interactively compete with homogeneous freezing in cirrus, we embedded the Ullrich et al. (2017) parameterizations in the Barahona and Nenes (2009) cirrus nucleation scheme. This scheme approximates the analytical solution of cloud parcel equations considering both homogeneous and heterogeneous nucleation and was ideal for use here because it allows the heterogeneous nucleation function to be customized.

For comprehensiveness, we implemented the Ullrich et al. (2017) parameterizations in both cirrus and mixed-phase clouds. We further implemented these parameterizations in both stratiform clouds and deep convective cores (using Morrison & Gettelman, 2008, and Song et al., 2012, microphysics, respectively). However, BC INPs in stratiform cirrus were found to dominate the total net radiative effects (see supporting information Table S1), so we primarily focus on impacts in these clouds. BC is now understood to be an ineffective INP at mixed-phase cloud temperatures (Mahrt et al., 2018; Vergara-Temprado et al., 2018). In convective cores, INPs may have opposite effects for moderate and strong convective systems (Zhao et al., 2019), while local nucleation at cirrus levels tends to be inhibited by an abundance of homogeneously frozen droplets lifted from below.

For each model configuration, we used four simulations to determine the impacts of anthropogenic BC INPs. A first pair of runs simulated PI and PD levels of BC with the abovedescribed setup, while a matching pair was run with all ice nucleation on BC turned off. We then calculated the ERF due to anthropogenic BC INPs while excluding other mechanisms (i.e., direct, semidirect, and warm cloud effects). This was done by comparing net cloud radiative effects between simulations as follows:

$$ERF_{anthroBCINPs} = (CRE_{PDBC} - CRE_{PIBC}) - (CRE_{PDBC}^{NoBCINPs} - CRE_{PIBC}^{NoBCINPs}).$$
(1)

Equivalent calculations were completed for SW and LW BC INP forcings, and all forcings were additionally calculated for cloud radiative effect changes in cirrus only. The cirrus forcings were calculated to better understand the response in cirrus and mirrored standard cloud radiative diagnostics while only considering clouds in grid cells colder than -38° C (based on the method in Gasparini & Lohmann, 2016). In addition to the core and sensitivity simulations, four additional simulations were ran to develop an understanding of diverse INP changes (Figure 2), and four more were used to separate BC effects by cloud type (Table S2). All simulation sets and stand-alone runs are briefly described in Table S1, and additional simulation details are presented in Text S1.



Geophysical Research Letters



Figure 2. Modeled changes to global cirrus radiative effects as INPs are added. Marked are the cirrus radiative effects before and after each INP addition, as arrow tails and heads, respectively. First, dust INPs are added with no other INPs present (dark yellow lines, initially 10% of the default number of dust INPs followed by the remaining 90%), then BC natural (green) and anthropogenic (blue) INPs are added, and lastly, all BC INP number concentrations are progressively increased to 10× the previous value (purple). Initial BC increases are represented twice, for simulations having the default amount of dust INPs (solid lines) and 10% as many (dashed). Background color shows net cirrus effects, being the sum of SW and LW effects. Black arrows serve as a guide to (i) the directions of positive and negative net forcings and (ii) the directions of cirrus radiative thickening or thinning equal in the shortwave and longwave (zero net forcing).

3. Results

3.1. Core Estimate of Anthropogenic BC INP Effect

Our core result is an estimate of the current ERF due to BC particles acting as INPs. Comparing simulations (see section 2) showed anthropogenic BC INPs to cause a net radiative forcing of -0.13 ± 0.07 W/m² globally and is expected to represent a cooling effect at the surface and lower troposphere. The presented uncertainty is the standard deviation among annual average forcings for 5 simulated years. This result did not factor in aircraft soot emissions, which we found to weaken the forcing to a small extent within the uncertainty range. In the next section we attribute the core result to a reduction in the simulated optical thickness of cirrus clouds, which generally warm the climate. Our result is shown in Figure 1e, in comparison with prior research, along with an uncertainty estimate from the range of sensitivity tests detailed below. Figure 1d shows BC INP impacts specifically on cirrus radiative effects, which we found to dominate the total radiative response. Impacts of BC INPs in mixed-phase clouds and deep convective cores were not found to be significant (see Table S2), and hence, we focus on stratiform cirrus for the remainder of this letter. We first describe how and why a moderate cooling impact occurs and then discuss to what extent this outcome is robust.

3.2. Process Understanding

The CESM2 simulations used here represent the chain of interactions from INP presence to associated radiative impacts. Depending on whether INPs optically thicken or thin clouds and whether this primarily impacts SW or LW radiation, INPs can cause either a warming or cooling effect. In our core simulations, anthropogenic BC impacts on net cirrus radiative effects (Figure 1d) roughly match the zonal structure of changes in homogeneous nucleation rates (Figure 1b) and high cloud optical depth (Figure 1c), suggesting a connection. Averaged globally, anthropogenic BC induces a decrease in cirrus nucleation rates and an optical thinning of cirrus. Impacts vary considerably by location, with both moderate cirrus thickening and thinning occurring (as seen in the percentile bounds in Figure 1c). Across latitudes, averaged anthropogenic BC INP effects on high cloud optical depth roughly correspond to changes in homogeneous (not heterogeneous) nucleation rates, showing the dominant influence of homogeneous nucleation with this model setup. In the southern tropics, radiative cooling is further enhanced by a BC INP-induced reduction in high cloud cover (Figure S2e).

BC INP impacts on cirrus radiative effects are the strongest, but also the most varied, in the tropics. SW effects are strong in this region partly because the underlying surface tends to be dark (Figure S2b), while LW effects are strong in part due to the cold cloud temperatures (and thus lower reemission of LW radiation to space) (Figure S2c). Local variations in these properties enable a change in cirrus optical depth to have highly varied effects depending on the cirrus affected. The tropics show the largest declines in total cirrus nucleation rates due to anthropogenic BC (Figure 1b). This is unsurprising since homogeneous nucleation is strongly prevalent in this region (Figure S1a) of frequent cirrus (Figure S2a), where convective events lift water, INPs, and liquid aerosol to high altitudes. These conditions provide an environment where modest shifts in the competition between homogeneous and heterogeneous nucleation can have large impacts. In a minority of high tropical cirrus, BC INPs counterintuitively cause homogeneous nucleation rates to increase. This is likely due to the BC INPs having secondary effects by influencing cirrus temperatures (Figure S2g) and updraft speeds (Figure S2h). The southern extratropics are relatively isolated from anthropogenic BC sources, and hence, cirrus thinning is relatively weak here. However, for these cirrus clouds, LW radiative effects tend to dominate—due in part to high underlying albedo (Figure S2b)—so their reduction does significantly alter the local radiative balance. In the northern extratropics, anthropogenic BC thins cirrus at nearly all locations despite the abundance of cirrus formed by heterogeneous nucleation, as the impact on homogeneous nucleation remains stronger. Hence, noticeable net cooling occurs at nearly all latitudes.

While our simulations show anthropogenic BC INPs to overall thin cirrus and cause cooling, the outcome depends greatly on the number and efficiency of these INPs as well as those of other colocated INPs, such as dust. Figure 2 shows the changes that occur to global-mean cirrus radiative effects as various INP additions are simulated. The addition of anthropogenic BC-shown in simulations both with default and low dust INP concentrations-causes a negative forcing by weakening cirrus LW effects more than SW effects, as is seen by comparing the slope of the arrows with the guidelines (and as $-\Delta LW_{cirr}/SW_{cirr} > 1$ in the accompanying table). However, as further BC INPs are added, the dominance of the LW impact diminishes (lowering $-\Delta LW_{cirr}/SW_{cirr}$). This weakens the net impact of each addition and eventually results in positive forcings when the SW weakening comes to dominate ($-\Delta LW_{cirr}/SW_{cirr} < 1$). Similarly, impacts are weakened by the presence of other colocated INP species, as shown by anthropogenic BC INPs causing nearly twice the forcing when dust INP number is reduced to 10% globally. Anthropogenic BC INPs induce a significant negative forcing because their emission sources allow them to reach some cirrus otherwise lacking sufficient INPs to saturate (or reverse) the effect. Mineral dust, the most recognized other INP, is primarily emitted from desert regions far from major anthropogenic sources and hence likely does not reach the same cirrus as anthropogenic BC. Additionally, BC tends to be smaller than dust and hence capable of being transported further without falling out. Once an initial amount of dust INPs are globally present, adding BC INPs has a substantially more LW dominant and net negative effect than adding further dust INPs.

Differences in affected homogeneously and heterogeneously formed cirrus may explain why the initially negative INP forcing diminishes and reverses as more INPs are added. Initial INPs have a relatively large impact on homogeneously formed cirrus radiative effects by preventing many freezing events per INP, inducing a negative forcing while the thickening effect in heterogeneously formed cirrus is too weak to be comparable. Gradually, as large-scale INP additions are made, homogeneously formed cirrus become less abundant, and their responses are outpaced by the net positive effect from thickening heterogeneously formed cirrus. Hastening this shift, changes to the thickness of heterogeneously formed cirrus, since cloud LW radiative effects saturate in thick cirrus over a range of optical depths where SW effects do not (Choi & Ho, 2006; Hong et al., 2016). Ultimately, thickening of heterogeneously formed cirrus results in a positive net effect even, while thinning of homogeneously formed cirrus may continue to dominate the globally averaged SW and LW responses (possibly explaining why in Figure 2 INP additions induce a cooling effect because they encounter and thin cirrus formed through homogeneous nucleation.





Figure 3. Global effective radiative forcings from anthropogenic BC INPs. Shown are outcomes from the control simulation set and all sensitivity tests. Uncertainties indicate one standard deviation among annual averages throughout the 5 year simulation period.

3.3. Sensitivity Tests

As model uncertainties can influence the balance between the warming and cooling effects of anthropogenic BC INPs, we ran sensitivity tests to determine the extent to which impacts could plausibly differ from our core result. ERFs from all sensitivity tests are shown in Figure 3. These tests represent a diverse range of plausible INP and cirrus characteristics. For one, BC INP number and efficiency are poorly constrained in simulated cirrus. Assuming a high bound on anthropogenic BC INP number concentrations of nearly 10× those in the default model (resembling the spread of middle-upper tropospheric BC mass among models; Samset et al., 2013; Schwarz et al., 2017) increased the global forcing by 16%, indicating a plausibly larger cooling effect but also that the impact of each INP added diminishes as more are added. Adding aircraft soot emissions (not otherwise simulated) affects cirrus primarily at Northern Hemisphere latitudes where heterogeneous nucleation already dominates, causing this INP source to primarily enhance cirrus SW and LW effects (Figure S4). While aircraft soot thus counterbalances the impacts of BC INPs on homogeneously formed cirrus with a slight positive net radiative impact, this source is not abundant enough to significantly weaken the anthropogenic BC INP forcing. Simulating BC INPs as if they all age into relatively inefficient INPs when emitted (1% as efficient as fresh) unsurprisingly rendered the forcing insignificant. More surprisingly, treating all BC as fresh INPs similarly reduced the net cooling effect by inducing SW thinning with little LW effect. For maximal cooling, INPs would cause cirrus to thin without substantially forming or thickening heterogeneously formed cirrus. Hence, regardless of possibilities for more numerous or effective BC INPs (largely equivalent) than assumed in the core finding, the narrow range of suitable INP concentrations for additional BC INPs to induce cooling (see Figure 2) considerably constrains the potential for an enhanced cooling. Similarly, a strong warming effect is difficult to achieve since it requires very high concentrations or efficiencies of BC INPs.

Anthropogenic BC INP impacts could hypothetically be stronger if the affected cirrus were different from those in our control simulation in ways that further enable BC INPs to thin cirrus more dominantly in the LW than SW. We tested this possibility by altering uncertain parameters that influence the balance between heterogeneous and homogeneous nucleation events. This includes updraft variability, which is crudely parameterized in the model, and the availability of both sulfate aerosol for homogeneous nucleation and additional INP species. Rerunning the model with an increased cirrus updraft spread enables a greater number of cirrus to reach the saturations necessary for homogeneous sulfate nucleation, increasing the frequency and extent of these events. In contrast, enhancing sulfate availability enabled more homogeneous

nucleation to occur only in the few cirrus where homogeneous nucleation is already strong enough to exhaust the supply of sulfate without the addition. These adjustments had opposite effects, with increased updraft variability resulting in a stronger negative forcing while increased sulfate availability reduced the forcing (+13% and -52% net forcings compared to the control, respectively). The difference is presumably because the sulfate increase thickens cirrus that are already LW saturated, while the updraft change enables the INPs to thin less thick cirrus. As seen in Table S3, the model setup with the updraft change has the strongest cirrus net radiative effect (+5.12 W/m², compared to +4.34 W/m² in the control) among sensitivity tests, while that with the sulfate change has the weakest (+3.83 W/m²). This follows the expectation that reducing the most warming cirrus would have the most cooling effect while thinning the least warming would cause less cooling. Similar sensitivity runs with the opposite changes (less updraft variability and less available sulfate) unsurprisingly led to weak INP radiative impacts. An additional test incorporated less numerous (10%) dust INPs in simulations. This roughly doubled the net anthropogenic BC forcing, being the highest enhancement among sensitivity tests and demonstrating the abovementioned concept of the strongest negative net effects when few INPs are present.

4. Conclusions

Our simulations showed PD anthropogenic emissions of BC aerosols to generally thin cirrus clouds through their role as INPs, resulting in a moderate net negative radiative forcing at the top of the atmosphere. Such a forcing is expected to cool Earth's surface and lower-middle troposphere, though—as seen in other studies of high altitude BC forcings (Samset & Myhre, 2015; Stjern et al., 2017)—the temperature response may not readily scale with the forcing. This study was unique in its consideration of BC INPs acting in diverse cloud types, which included stratiform and convective clouds at both mixed-phase and cirrus temperatures. Convective clouds in particular have been ignored in previous studies. The use of contemporary laboratory fittings for BC ice nucleating ability showed BC to be a weak INP at mixed-phase temperatures with negligible net radiative impacts, a result that disagrees with the consensus of a warming BC INP effect in these clouds (Bond et al., 2013) (see Figure 1e). Overall, anthropogenic BC INPs caused a negative forcing that was primarily due to their presence in stratiform cirrus within environments that in their absence were capable of supporting homogeneous freezing. Our result of a small cooling in stratiform cirrus resembled findings in some studies (Gettelman et al., 2012; Penner et al., 2018) though in bounding this estimate we diverged from the consensus that a warming effect is as likely (Bond et al., 2013) (Figure 1e).

Among tests representing a range of plausible heterogeneous and homogeneous ice nucleation strengths and sensitivities to BC INPs, anthropogenic BC INPs were found to have an ERF ranging from -0.30 to +0.02 W/m². The methods used to establish this range did have some limitations that we note here. While we tested BC INP effects for a comprehensive range of realizations, plausible outcomes may not have been entirely constrained due to the reliance of sensitivity tests on one model and uncertain parameter bounds. All simulations had cirrus radiative effects (Table S3) well within the global mean observational bounds of Hong et al. (2016), indicating potentially more uncertainty in cirrus properties than tested. As with other studies that simulated BC INP effects without observations to directly corroborate findings, confidence in the stated bounds remains low. We did not incorporate the possibility that BC aged through cloud processing may substantially enhance its INP efficiency (Mahrt et al., 2020), which could either strengthen or reduce the forcing. Conversely, as many types of BC have been found in laboratory experiments to be ineffective INPs (Mahrt et al., 2018) and BC at cirrus levels may not match the samples used to create parameterizations, there is a possibility that BC is instead a weaker INP than represented here. Lastly, while simulations tested BC INP effects in a comprehensive range of ice clouds, they relied on a single model treatment of ice optical properties and some clouds were likely not well represented due to the coarse model resolution.

We expect that the stated range represents a realistic bound on the anthropogenic BC INP effect despite the mentioned limitations. The identified negative forcing depends on the existence of a limited range of cirrus types, neither too depleted of homogeneously formed crystals nor too thick for thinning to cool. A substantially greater negative forcing appears unlikely, as the driving effect becomes saturated with sufficient INP presence. A sizable positive forcing may also be improbable, as this was shown to necessitate orders of magnitude more INPs than in the core model setup or an equivalently strong shift toward more heterogeneous and less homogeneous nucleation. With simulations incorporating physical and laboratory-derived



equations of ice nucleation rates over a range of plausible inputs (INPs, liquid aerosol, and updrafts), no such situation is identified as feasible. By contrast, all but one sensitivity test evidence a small to moderate negative anthropogenic BC INP forcing. The forcing could potentially negate some or all of the positive forcing caused by BC direct effects, variously estimated as $0.08-1.27 \text{ W/m}^2$ (Bond et al., 2013) and $0.16-1.40 \text{ W/m}^2$ (Wang et al., 2016). This could further diminish the climate mitigation potential from decreases in BC emissions (Lund et al., 2020) and may have ramifications for climate policy.

Data Availability Statement

Averaged output from model simulations used is available online (http://doi.org/10.5281/zenodo.3863025).

References

- Baker, L., Collins, W. J., Olivié, D. J. L., Cherian, R., Myhre, G., & Quaas, J. (2015). Climate responses to anthropogenic emissions of short-lived climate pollutants. *Atmospheric Chemistry and Physics*, 15(14), 8201–8216. https://doi.org/10.5194/acp-15-8201-2015
 Barahona, D., & Nenes, A. (2009). Parameterizing the competition between homogeneous and heterogeneous freezing in ice cloud
- formation-polydisperse ice nuclei. *Atmospheric Chemistry and Physics*, 9(16), 5933–5948. https://doi.org/10.5194/acp-9-5933-2009 Bond, T. C., Doherty, S. J., Fahey, D. W., Forster, P. M., Berntsen, T., DeAngelo, B. J., et al. (2013). Bounding the role of black carbon in the
- climate system: A scientific assessment. Journal of Geophysical Research: Atmospheres, 118, 5380–5552. https://doi.org/10.1002/jgrd.50171
- Boucher, O., Balkanski, Y., Hodnebrog, Ø., Myhre, C. L., Myhre, G., Quaas, J., et al. (2016). Jury is still out on the radiative forcing by black carbon. Proceedings of the National Academy of Sciences, 113(35), E5092–E5093. https://doi.org/10.1073/pnas.1607005113
- Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P., et al. (2013). Clouds and aerosols. In Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 571–657). Cambridge, UK, and New York, NY: Cambridge University Press.
- Choi, Y. S., & Ho, C. H. (2006). Radiative effect of cirrus with different optical properties over the tropics in MODIS and CERES observations. Geophysical Research Letters, 33, L21811. https://doi.org/10.1029/2006GL027403
- Corti, T., & Peter, T. (2009). A simple model for cloud radiative forcing. Atmospheric Chemistry and Physics, 9(15), 5751–5758. https://doi.org/10.5194/acp-9-5751-2009

Danabasoglu, G., Lamarque, J. F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The Community Earth System Model version 2 (CESM2). *Journal of Advances in Modeling Earth Systems*, 12, e2019MS001916. https://doi.org/10.1029/2019MS001916 Gasparini, B., & Lohmann, U. (2016). Why cirrus cloud seeding cannot substantially cool the planet. *Journal of Geophysical Research:*

- Atmosphere, 121(9), 4877–4893. https://doi.org/10.1002/2015JD024666 Gettelman, A., Liu, X., Barahona, D., Lohmann, U., & Chen, C. (2012). Climate impacts of ice nucleation. Journal of Geophysical Research, 117, D20201. https://doi.org/10.1029/2012JD017950
- Hoesly, R. M., Smith, S. J., Feng, L., Klimont, Z., Janssens-Maenhout, G., Pitkanen, T., et al. (2018). Historical (1750-2014) anthropogenic emissions of reactive gases and aerosols from the Community Emission Data System (CEDS). *Geoscientific Model Development*, 11, 369–408. https://doi.org/10.5194/gmd-11-369-2018
- Hong, Y., Liu, G., & Li, J. L. (2016). Assessing the radiative effects of global ice clouds based on CloudSat and CALIPSO measurements. Journal of Climate, 29, 7651–7674. https://doi.org/10.1175/JCLI-D-15-0799.1
- Hoose, C., & Möhler, O. (2012). Heterogeneous ice nucleation on atmospheric aerosols: A review of results from laboratory experiments. Atmospheric Chemistry and Physics, 74(3), 699–717. https://doi.org/10.1175/JAS-D-16-0074.1
- Kanji, Z. A., Ladino, L. A., Wex, H., Boose, Y., Burkert-Kohn, M., Cziczo, D. J., & Krämer, M. (2017). Overview of ice nucleating particles. *Meteorological Monographs*, 58, 1.1–1.33. https://doi.org/10.1175/AMSMONOGRAPHS-D-16-0006.1
- Liu, X., Ma, P. L., Wang, H., Tilmes, S., Singh, B., Easter, R. C., et al. (2016). Description and evaluation of a new four-mode version of the Modal Aerosol Module (MAM4) within version 5.3 of the Community Atmosphere Model. *Geoscientific Model Development*, 9(PNNL-SA-110649(2)), 505–522. https://doi.org/10.5194/gmd-9-505-2016
- Lund, M. T., Aamaas, B., Stjern, C. W., Klimont, Z., Berntsen, T. K., & Samset, B. H. (2020). A continued role of short-lived climate forcers under the shared socioeconomic pathways. *Earth System Dynamics Discussions*, 1–23. https://doi.org/10.5194/esd-2020-9
- Lund, M. T., Samset, B. H., Skeie, R. B., Watson-Parris, D., Katich, J. M., Schwarz, J. P., & Weinzierl, B. (2018). Short black carbon ifetime inferred from a global set of aircraft observations. npj Climate and Atmospheric Science, 1, 1–8. https://doi.org/10.1038/ s41612-018-0040-x
- Mahrt, F., Kilchhofer, K., Marcolli, C., Grönquist, P., David, R. O., Rösch, M., et al. (2020). The impact of cloud processing on the ice nucleation abilities of soot particles at cirrus temperatures. *Journal of Geophysical Research: Atmospheres*, 125, e2019JD030922. https:// doi.org/10.1029/2019JD030922
- Mahrt, F., Marcolli, C., David, R. O., Grönquist, P., Barthazy Meier, E. J., Lohmann, U., & Kanji, Z. A. (2018). Ice nucleation abilities of soot particles determined with the horizontal ice nucleation chamber. Atmospheric Chemistry and Physics, 18(18), 13,363–13,392. https://doi. org/10.5194/acp-18-13363-2018
- Morrison, H., & Gettelman, A. (2008). A new two-moment bulk stratiform cloud microphysics scheme in the Community Atmosphere Model, version 3 (CAM3). Part I: Description and numerical tests. *Journal of Climate*, 21, 3642–3659. https://doi.org/10.1175/ 2008JCLI2105.1
- Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., & Huang, J. (2013). Anthropogenic and natural radiative forcing. In *Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 659–740). United Kingdom and New York, NY: Cambridge University Press.
- Peng, J., Hu, M., Guo, S., Du, Z., Zheng, J., Shang, D., et al. (2016). Markedly enhanced absorption and direct radiative forcing of black carbon under polluted urban environments. *Proceedings of the National Academy of Sciences*, 113(16), 4266–4271. https://doi.org/ 10.1073/pnas.1602310113
- Penner, J. E., Zhou, C., Garnier, A., & Mitchell, D. L. (2018). Anthropogenic aerosol indirect effects in cirrus clouds. *Journal of Geophysical Research: Atmospheres*, 123(20). https://doi.org/10.1029/2018jd029204

Acknowledgments

This research was supported by grants #295046 from the Research Council of Norway and #758005 from the European Research Council. The simulations were performed on resources provided by UNINETT Sigma2-the National Infrastructure for High Performance Computing and Data Storage in Norway. C. W. S. and B. H. S. acknowledge funding by the Research Council of Norway through the grant NetBC (244141).

- Samset, B. H., & Myhre, G. (2015). Climate response to externally mixed black carbon as a function of altitude. Journal of Geophysical Research: Atmospheres, 120, 2913–2927. https://doi.org/10.1002/2014JD022849
- Samset, B. H., Myhre, G., Forster, P. M., Hodnebrog, Ø., Andrews, T., Faluvegi, G., et al. (2016). Fast and slow precipitation responses to individual climate forcers: A PDRMIP multimodel study. *Geophysical Research Letters*, 43, 2782–2791. https://doi.org/10.1002/ 2016GL068064
- Samset, B. H., Myhre, G., Schulz, M., Balkanski, Y., Bauer, S., Berntsen, T. K., et al. (2013). Black carbon vertical profiles strongly affect its radiative forcing uncertainty. *Atmospheric Chemistry and Physics*, *13*(5), 2423–2434. https://doi.org/10.5194/acp-13-2423-2013
- Schwarz, J. P., Weinzierl, B., Samset, B. H., Dollner, M., Heimerl, K., Markovic, M. Z., et al. (2017). Aircraft measurements of black carbon vertical profiles show upper tropospheric variability and stability. *Geophysical Research Letters*, 44, 1132–1140. https://doi.org/10.1002/ 2016GL071241
- Song, X., Zhang, G. J., & Li, J. L. (2012). Evaluation of microphysics parameterization for convective clouds in the NCAR Community Atmosphere Model CAM5. Journal of Climate, 25, 8568–8590. https://doi.org/10.1175/JCLI-D-11-00563.1
- Stjern, C. W., Samset, B. H., Myhre, G., Forster, P. M., Hodnebrog, Ø., Andrews, T., et al. (2017). Rapid adjustments cause weak surface temperature response to increased black carbon concentrations. *Journal of Geophysical Research: Atmospheres*, 122, 11,462–11,481. https://doi.org/10.1002/2017JD027326
- Sullivan, S. C., Morales Betancourt, R., Barahona, D., & Nenes, A. (2016). Understanding cirrus ice crystal number variability for different heterogeneous ice nucleation spectra. Atmospheric Chemistry and Physics, 16, 2611–2629. https://doi.org/10.5194/acp-16-2611-2016
- Ullrich, R., Hoose, C., Cziczo, D. J., Froyd, K. D., Schwarz, J. P., Perring, A. E., et al. (2019). Comparison of modeled and measured ice nucleating particle composition in a cirrus cloud. *Journal of the Atmospheric Sciences*, 76(4), 1015–1029. https://doi.org/10.1175/JAS-D-18-0034.1
- Ullrich, R., Hoose, C., Möhler, O., Niemand, M., Wagner, R., Höhler, K., et al. (2017). A new ice nucleation active site parameterization for desert dust and soot. *Journal of the Atmospheric Sciences*, 74(3), 699–717. https://doi.org/10.1175/JAS-D-16-0074.1
- Vergara-Temprado, J., Holden, M. A., Orton, T. R., O'Sullivan, D., Umo, N. S., Browse, J., et al. (2018). Is black carbon an unimportant ice-nucleating particle in mixed-phase clouds? *Journal of Geophysical Research: Atmospheres*, 123, 4273–4283. https://doi.org/10.1002/ 2017JD027831
- Wang, R., Balkanski, Y., Boucher, O., Ciais, P., Schuster, G. L., Chevallier, F., et al. (2016). Estimation of global black carbon direct radiative forcing and its uncertainty constrained by observations. *Journal of Geophysical Research: Atmospheres*, 121, 5948–5971. https://doi.org/ 10.1002/2015JD024326
- Zhao, B., Gu, Y., Liou, K.-N., Wang, Y., Liu, X., Huang, L., et al. (2018). Type-dependent responses of ice cloud properties to aerosols from satellite retrievals. *Geophysical Research Letters*, 45, 3297–3306. https://doi.org/10.1002/2018GL077261
- Zhao, B., Wang, Y., Gu, Y., Liou, K. N., Jiang, J. H., Fan, J., et al. (2019). Ice nucleation by aerosols from anthropogenic pollution. *Nature Geoscience*, *12*(8), 602–607. https://doi.org/10.1038/s41561-019-0389-4

References From the Supporting Information

- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. Quarterly Journal of the Royal Meteorological Society, 137(656), 553–597. https://doi.org/ 10.1002/qj.828
- Liu, X., & Penner, J. E. (2005). Ice nucleation parameterization for global models. *Meteorologische Zeitschrift*, 14(4), 499–514. https://doi. org/10.1127/0941-2948/2005/0059
- Loeb, N. G., Doelling, D. R., Wang, H., Su, W., Nguyen, C., Corbett, J. G., et al. (2018). Clouds and the Earth's Radiant Energy System (CERES) energy balanced and filled (EBAF) top-of-atmosphere (TOA) edition-4.0 data product. *Journal of Climate*, *31*(2), 895–918. https://doi.org/10.1175/JCLI-D-17-0208.1
- Sourdeval, O., Gryspeerdt, E., Krämer, M., Goren, T., Delanoë, J., Afchine, A., et al. (2018). (2018). Ice crystal number concentration estimates from lidar-radar satellite remote sensing—Part 1: Method and evaluation. *Atmospheric Chemistry and Physics*, 18(19), 14,327–14,350. https://dx.doi.org/10.5194/acp-18-14327-2018